 Vc/9-Nov-04
Week46 starting 8Nov 04

SM18 Operation Meeting (Tuesdays)

 Bench status  on Monday 8 Nov 04  ~10.00 hrs 

	LHC SM18 Magnet Test Facility 


	4
	A1
	MBBL2098
	PT 4 Slow power abort check
	1.9

	1
	A2
	MBBL3219
	PT 12 Warm Up
	116

	6
	B1
	SSS042
	Prep 5 Pumping + Cool Down
	295.6

	12
	B2
	SSS069
	ICS 5 Traveller Last
	297.4

	5
	C1
	MBBR3243
	Prep 5 Pumping + Cool Down
	148.6

	9
	C2
	_
	_
	296.3

	3
	D1
	SSS075
	PT 8 Fast De-excitation
	1.9

	10
	D2
	SSS071
	CR 1 Leak Test
	295.9

	2
	E1
	MBBL2116
	PT 6.2 Training 2
	5.2

	8
	E2
	MBAL2097
	ICS 3 Traveller after warm test
	295.8

	7
	F1
	MBAL2092
	ICS 4 Final connection
	294.5

	11
	F2
	MBBL3221
	ICS 5 Traveller Last
	296.9


11/8/2004 9:50:45 AM    Temperature data are taken from TBSM18 site
Coordinator this week: Gijs De Rijk; MTM support on call :  (16-3602 Maryline).

14 Magnets:

 9 dipoles( 0 with MM)+ 5 SSS(1 with MM) +  0 SSSS ]   Disconnected after tests last week ( Sat Midnight to Sat midnight) 
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
Traveller & SSS & Warm SSW
AOB

	Magnets under Test from 31-10-2004 to 06-11-2004 

	-
	Bench
	Magnet
	Shafts
	Start Date .
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	End Date
	Total
	
	

	1
	TBA1
	MBBL1149 
	[image: image1.png]



	27-10-2004 
	21 h.
	25 h.
	13 h.
	18 h.
	7 h.
	31-10-2004 
	96 hours
	
	

	2
	TBB1
	SSS067 
	[image: image2.png]



	27-10-2004 
	38 h.
	38 h.
	65 h.
	14 h.
	7 h.
	03-11-2004 
	168 hours
	
	

	3
	TBB2
	SSS050 
	[image: image3.png]



	25-10-2004 
	43 h.
	24 h.
	56 h.
	12 h.
	14 h.
	01-11-2004 
	168 hours
	
	

	4
	TBC1
	MBBL2115 
	[image: image4.png]



	28-10-2004 
	32 h.
	29 h.
	17 h.
	18 h.
	8 h.
	01-11-2004 
	96 hours
	
	

	5
	TBC1
	MBBR3222 
	[image: image5.png]



	01-11-2004 
	24 h.
	43 h.
	17 h.
	15 h.
	5 h.
	06-11-2004 
	120 hours
	
	

	6
	TBC2
	MBBL1163 
	[image: image6.png]



	30-10-2004 
	24 h.
	37 h.
	17 h.
	19 h.
	5 h.
	04-11-2004 
	120 hours
	
	

	7
	TBD1
	SSS014 
	[image: image7.png]



	21-10-2004 
	115 h.
	32 h.
	91 h.
	10 h.
	9 h.
	31-10-2004 
	240 hours
	
	

	8
	TBD1
	SSS027 
	[image: image8.png]



	31-10-2004 
	35 h.
	27 h.
	44 h.
	8 h.
	9 h.
	05-11-2004 
	120 hours
	
	

	9
	TBD2
	SSS065 
	[image: image9.png]



	24-10-2004 
	61 h.
	96 h.
	62 h.
	14 h.
	14 h.
	03-11-2004 
	240 hours
	
	

	10
	TBE1
	MBAL1094 
	[image: image10.png]



	31-10-2004 
	30 h.
	30 h.
	17 h.
	14 h.
	6 h.
	05-11-2004 
	120 hours
	
	

	11
	TBE1
	MBAL1157 
	[image: image11.png]



	27-10-2004 
	26 h.
	40 h.
	20 h.
	13 h.
	10 h.
	31-10-2004 
	96 hours
	
	

	12
	TBE2
	MBBL3099 
	[image: image12.png]



	29-10-2004 
	42 h.
	49 h.
	12 h.
	13 h.
	6 h.
	03-11-2004 
	120 hours
	
	

	13
	TBF1
	MBAL1160 
	[image: image13.png]



	27-10-2004 
	27 h.
	29 h.
	17 h.
	25 h.
	8 h.
	01-11-2004 
	120 hours
	
	

	14
	TBF2
	MBBR3188 
	[image: image14.png]



	28-10-2004 
	25 h.
	44 h.
	16 h.
	17 h.
	10 h.
	01-11-2004 
	96 hours
	
	


Quench Information
 

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBBL1149
	12300 A.
(D1-L)
	12706 A.
(D1_U)
	[image: image15.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	SSS067
	11783 A.
(QE-34)
	12848 A.
(QI-12)
	[image: image16.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	SSS050
	12200 A.
(QI-34)
	12732 A.
(QE-34)
	[image: image17.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	MBBL2115
	10700 A.
(D2-U)
	12100 A.
(D1-L)
	[image: image18.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	MBBR3222
	12057 A.
(D2-U)
	12469 A.
(D2_U)
	[image: image19.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	MBBL1163
	12042 A.
(D1_U)
	12541 A.
(D2_L)
	[image: image20.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	SSS014
	12137 A.
(QI-34)
	12314 A.
(QE-12)
	[image: image21.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	SSS027
	11700 A.
(QE_12)
	12498 A.
(QE_34)
	[image: image22.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	SSS065
	11653 A.
(QE-12)
	12553 A.
(QI-12)
	[image: image23.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	MBAL1094
	12195 A.
(D1-U)
	12835 A.
(D2-L)
	[image: image24.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBAL1157
	11724 A.
(D1_U)
	12420 A.
(D1-U)
	[image: image25.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	MBBL3099
	12055 A.
(D2_L)
	12112 A.
(D2-U)
	[image: image26.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	13
	MBAL1160
	11511 A.
(No Data)
	12483 A.
(D1_L)
	[image: image27.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	14
	MBBR3188
	12307 A.
(D1_U)
	12383 A.
(D1-U)
	[image: image28.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-


	From 31-10-2004 to 06-11-2004 

	-
	Bench
	Magnet Name
	shafts
	Start Date .
	Connect
Time
	Wait
+Cooling
	Cold
Tests
	Wait 
+Warming
	Disconnect
Time
	End Date .
	Total
	Q

	1
	TBF2
	MBBR3188 
	False 
	28-10-2004 
	25 h.
	44 h.
	16 h.
	17 h.
	10 h.
	01-11-2004 
	4 days
	2 

	2
	TBF1
	MBAL1160 
	False 
	27-10-2004 
	27 h.
	29 h.
	17 h.
	25 h.
	8 h.
	01-11-2004 
	5 days
	2 

	3
	TBE2
	MBBL3099 
	False 
	29-10-2004 
	42 h.
	49 h.
	12 h.
	13 h.
	6 h.
	03-11-2004 
	5 days
	2 

	4
	TBE1
	MBAL1094 
	False 
	31-10-2004 
	30 h.
	30 h.
	17 h.
	14 h.
	6 h.
	05-11-2004 
	5 days
	2 

	5
	TBE1
	MBAL1157 
	False 
	27-10-2004 
	26 h.
	40 h.
	20 h.
	13 h.
	10 h.
	31-10-2004 
	4 days
	2 

	6
	TBD2
	SSS065 
	False 
	24-10-2004 
	61 h.
	96 h.
	62 h.
	14 h.
	14 h.
	03-11-2004 
	10 days
	2 

	7
	TBD1
	SSS014 
	True 
	21-10-2004 
	115 h.
	32 h.
	91 h.
	10 h.
	9 h.
	31-10-2004 
	10 days
	2 

	8
	TBD1
	SSS027 
	False 
	31-10-2004 
	35 h.
	27 h.
	44 h.
	8 h.
	9 h.
	05-11-2004 
	5 days
	2 

	9
	TBC2
	MBBL1163 
	False 
	30-10-2004 
	24 h.
	37 h.
	17 h.
	19 h.
	5 h.
	04-11-2004 
	5 days
	2 

	10
	TBC1
	MBBL2115 
	False 
	28-10-2004 
	32 h.
	29 h.
	17 h.
	18 h.
	8 h.
	01-11-2004 
	4 days
	2 

	11
	TBC1
	MBBR3222 
	False 
	01-11-2004 
	24 h.
	43 h.
	17 h.
	15 h.
	5 h.
	06-11-2004 
	5 days
	2 

	12
	TBB2
	SSS050 
	False 
	25-10-2004 
	43 h.
	24 h.
	56 h.
	12 h.
	14 h.
	01-11-2004 
	7 days
	2 

	13
	TBB1
	SSS067 
	False 
	27-10-2004 
	38 h.
	38 h.
	65 h.
	14 h.
	7 h.
	03-11-2004 
	7 days
	2 

	14
	TBA1
	MBBL1149 
	False 
	27-10-2004 
	21 h.
	25 h.
	13 h.
	18 h.
	7 h.
	31-10-2004 
	4 days
	2 


	

	
	
	


from 31-10-2004 to 06-11-2004
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PROBLEMS OF LAST WEEK

Problem chronology  of the SSW Measurements done at Cold for SSS 027 on TBD1.
      Test Done                                                    No of times Repeated                          Date and time                                               Status
SC21:Installation of SSW in AP 2                                   1                                                    3/11/04         9:30 PM                                   OK

FS21: Field Strength Measurement in AP 2 @ 760 A       4                                                   4/11/04         00.20 AM                            Failed
FS22: Field Strength Measurement in AP 2 @ 5000 A     3                                                   4/11/04         00:53 AM                            Failed
FS23:Field Strength Measurement in AP 2 @ 5000 A       2                                                   4/11/04         3:02 AM                              Failed
SC22:Roll angle Measurement in AP2 - Quadrapole          1                                                   4/11/04         5:29 AM                              Failed
                                                                                                                                                                                         (Wire broke and it is reinstalled)

SC21:Installation of SSW in AP 2                                   1                                                   4/11/04         3:30 PM                               OK

SC22:Roll angle Measurement in AP2 - Quadrapole         7                                                    4/11/04         3:40 PM                               Failed
                                                                                                                                                                                                     (Uncertainty very high)
SC11:Installation of SSW in AP 1                                1                                                    4/11/04         7:00 PM                              OK

SC12:Roll angle Measurement in AP1 - Quadrapole        1                                                     4/11/04         7:10 PM                              Failed
                                                                                                                                                                                            (Readings out of range)
	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday _02_N


	TBD2
	CDW2 on SSS065: resistance values of sextupoles are out of ranges. Already observed before cold tests


	MTM
	
	Stripping not given before investigation.

	Tuesday _02_N


	TBB1
	IAP at warm final: results in bothMQAX-R67 and MQXX-R67.


	AB/CO
	IAP
	Results merged in the proper folder.

	Wednesday _03_A


	TBA2
	LHC cycles on 3219 magnet. Angles between forward and downward rotation too high in the current ramp.


	MTM
	MMP
	Stephane contacted. The values measured during the current ramp are not correct We should consider only those on current plateau. After check he advised to proceed. 



	Wednesday_03_N
	TBD1
	SSW measurements on SSS065. Motor blocked. 

Bad results obtained for strength measurements at cold.

Wire broken.
	MTM
	
	Test stopped and re-lauunched.

Check by N.Smirnov.

	Wednesday_03_N
	
	Rocla blocked in front of the C bench
	
	
	9 hours of stop.

	Wednesday_03_N

Sunday_07_A
	TBF1

TBF2
	Wrong HV test sequence performed at cold : The sequence of HV at warm was applied.
	MTM
	
	Error found Test with sequence at cold  re-launched.

	Wednesday_03_N
	TBA2
	Problem in the scan of the long anti-cryostat occurring in the middle of the cold test (after 6 LHC cycles)?
	MTM
	
	Anti-cryostat coefficients recognized now for the rest of the test.

Maryline will trace back the problem.

	Thursday_04_M
	
	PT4: Communication problem between PLC and PC
	AB/CO?
	
	

	Thursday_04_M
	
	PT5.1: Data not visible in TEMA .Analysis done in local.
	AB/CO?
	
	

	Thursday_04_M
	TBF1
	Connector of the cable connecting the TRU and the mobile rack is found damaged
	MTM
	
	

	Thursday_04_M
	TBD1
	SPA occurred during roll angle measurements.

Roll angle values and uncertainty out of the limits
	MTM
	
	

	Thursday_04_A
	TBD2
	HV test failed on SSS071

Resistance value out of range for Ext.dipole and MQT.
	CRI
	HV
	Richard called. Investigation 2 non conformities opened.

-VTaps inverted in the MQT: proceed as is.

-Inversion of the Vtaps of the dipole performed by Richard.

	Thursday_04_A
	TBD2
	SSW at cold giving bad results. Good at warm
	
	
	Nicolay, Guy contacted. Software problem. J.Di Marco contacted. Still unsolved

	Thursday_04_N
	TBD1
	HV test launched with XMT.MY95 rack. Communication problem. 
	AB/CO?
	HV
	Rack changed. OK. Richard informed.

	Friday_05_M
	TBF1
	LHC cycle not inserted during the training. 
	MTM
	
	TC informed. Provoked quench at 10 kA and test done.

	Friday_05_M
	TBA2
	SPA during LHC cycle at 11850 A 
	?
	
	

	Friday_05_M
	TBF1 
	SPA during LHC cycle: DCCT falls down. 
	AB/PO
	
	Richard put the DCCT on. Quench performed and LHC done

	Friday_05_A
	TBF1 
	Fault on the motor during LHC cycle.
	MTM
	
	Cable rolled around the motor axis. Richard and Guy solved the problem.

	Friday_05_A
	TBB2
	TEMA not responding. Sun crashed.
	AB/CO
	
	Richard investigated. Power supply of the sun out of work. Richard phoned to Magnus and Alessandro. Richard took the power supply of the SUN in the cluster G and replaced the dead one in that of the cluster B. 

	Friday_05_N
	TBB2
	Permanent trigger coming from the chassis 4 of the Potaim-cards.
	MTM
	
	Richard called. He found a problem with one connection cable. He decide to un-validate the concerned card. He will see P.Legrand Monday to check the cable.

	Saturday_06_M
	TBD1 
	Lyra test: not a good value of Vmagnet at warm. (2.68 V instead of 3.54 V)
	ICS
	
	Richard suspected  a problem in the VTaps soldered by ICS. He put a strapping in the chassis to by pass these to taps. Then OK for Lyra test.

	Saturday_06_A
	TBC1
	One Vtaps cut in the CFB
	MTM
	
	Richard informed and by pass it in the chassis. Pot-Aim.

	Sunday 07_A
	TBD2
	SSS071: leak test failed.
	
	
	Problem coming from the line X?  leak on the MRB? TC informed.

Disconnection of the magnet decide Monday..

	Sunday 07_A
	TBF2
	CDAP HV : Warm sequence followed (voltage up to 600 V).
	MTM
	HV
	Test repeated with the cold sequence.

	Sunday 07_A
	TBB1
	IAP at warm: resistance measurements not correct.
	MTM
	IAP
	Richard called. SSS type B-F. Adaptor needed plugged on the connector P11 to redistribute the V taps.











